Author’s response to reviews

Title: Parents and nurses telling their stories: The perceived needs of parents caring for critically ill children at Kilimanjaro Christian Medical Centre in Tanzania

Authors:

Vivian Frank Saria (vivianfrank@gmail.com)

Lilian Teddy Mselle (nakutz@yahoo.com)

Birgit Anne Siceloff (bjorgensen59@gmail.com)

Version: 1 Date: 14 Jan 2019

Author’s response to reviews:

The discussion section has been revised as suggested by the editor and the typos corrected accordingly.